Author’s response to reviews

Title: Electrical Conductivity-based Contrast Imaging for Characterizing Prostatic Tissues: In Vivo Animal Feasibility Study

Authors:
Yong Soo Cho (waveys1@hanmail.net)
Young Hoe Hur (surgihur@naver.com)
Hyun Ju Seon (sunaura@hanmail.net)
Jin Woong Kim (jw4249@hanmail.net)
Hyung Joong Kim (bmekim@khu.ac.kr)

Version: 3 Date: 07 Oct 2019

Author’s response to reviews:

1. Ethical approval
Please confirm in the "Ethics approval and consent to participate" section of your Declarations that the named ethics committee approved the study, as you have indicated clearly in your Methods section.
= We improved the sentence in the "Method" and "Ethics approval and consent to participate" sections.

2. Euthanasia
Please clarify in the Methods section if the beagles used in the study were euthanised, or if they were not (if this is the case, please clarify what was done with the beagles after the study was complete). If they were euthanised, clarify your euthanasia methods, including whether animals were anaesthetised and/or unconscious, injection dosages if applicable, methods used and rationale etc. Please try to be as detailed as possible.
= We added a sentence and cite a reference [19] in the "Methods" subsection "Imaging Experiment".

3. Role of Funders
Please note the role(s) of the funding body (bodies) in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript, which should be declared in the Funding section.
= We added a sentence for this.

4. Remove supplementary file
Please remove the supplementary files "Details of revision.doc" and the NC3R ARRIVE guidelines checklist from the manuscript, as they are not required to be attached to the manuscript at this stage of the editorial process.
5. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

= We made correction as your comment.